Author's response to reviews

Title: Neuropsychological effects of chronic low-dose exposure to polychlorinated biphenyls (PCBs): A cross-sectional study

Authors:

Martin Peper (peper@psychologie.uni-freiburg.de)
Martin Klett (martin.klett@rhein-neckar-kreis.de)
Rudolf Morgenstern (rudolf.morgenstern@charite.de)

Version: 6 Date: 17 October 2005

Author's response to reviews: see over
Re: Ms Martin Peper, Martin Klett, and Rudolf Morgenstern: Neuropsychological effects of chronic low-dose exposure to polychlorinated biphenyls (PCB): A cross-sectional study

Dear Dr. Grandjean,

- I agree with the solution chosen for the title page. If you agree, the second address should then be given in the "Corresponding author" line. I added my address in this line.

- We have examined the reference list and have made some corrections for the first thirty or so; cf. attached file. Please take a look at these corrections and apply the same formatting principles in regard to the remaining references. Formatting and correction was done.

- We have also indicated some minor changes, e.g., on page 1. I did not find this; however, I once again looked through the ms and corrected some minor errors.

- Table 4 will not fit the journal page in portrait format. Either the table should be redesigned with fewer columns, or it should be uploaded as a separate file. I compressed and shortened Table 4 once again. It is now smaller than Table 3 which appeared to fit the Journal page. Thus, Table 4 should now be prepared to be included in the ms.

Yours sincerely,

Martin Peper, PhD MD